**Author's response to reviews**

**Title:** Dietary differences between elderly Iranians living in Sweden and Iran A cross-sectional comparative study

**Authors:**

Afsaneh Koochek (Afsanek.Koochek@ikv.uu.se)  
Parvin Mirmiran (mirmiran@erc.ac.ir)  
Kristina Sundquist (Kristina.Sundquist@ki.se)  
Firoozeh Hosseini (firoozehhosseini@yahoo.com)  
Tohid Azizi (webmaster@endocrine.ac.ir)  
Ali S Moeini (Moeini@erc.ac.ir)  
Sven-Erik Johansson (sven-erik.johansson@ki.se)  
Brita Karlström (Brita.Karlstrom@ikv.uu.se)  
Fereidoun Azizi (azizi@erc.ac.ir)  
Jan Sundquist (Jan.Sundquist@med.lu.se)

**Version:** 4  **Date:** 16 May 2011

**Author's response to reviews:**

I will keep the most recent version of cover letter which was uploaded 22 January 2011, because the manuscript only needed formatting changes according to recommendations from The BMC Series Editorial Production Team.